Title: Availability and quality of paraffin blocks identified in pathology archives: A multi-institutional study by the Shared Pathology Informatics Network (SPIN)

Authors:

Ashkokkumar A Patel (patelaa@upmc.edu)  
Dilipkumar Gupta (dkgupta10@hotmail.com)  
David Seligson (dseligson@mednet.ucla.edu)  
Eyas M Hattab (ehattab@iupui.edu)  
Ulysses J Balis (ubalis@partners.org)  
Thomas M Ulbright (tulbrigh@iupui.edu)  
Isaac S Kohane (isaac_kohane@harvard.edu)  
Jules J Berman (jjberman@alum.mit.edu)  
John R Gilbertson (jgi@alum.mit.edu)  
Sarah Dry (sdry@mednet.ucla.edu)  
Osvaldo Schirripa (oschirripa@mednet.ucla.edu)  
Hong Yu (hyu@mednet.ucla.edu)  
Michael J Becich (becich@pitt.edu)  
Anil V Parwani (parwaniav@upmc.edu)  
Shared Pathology Informatics Network (spin) (spin@pop.nci.nih.gov)

Version: 7 Date: 27 February 2007

Author's response to reviews: see over
February 27, 2007

Dear Editorial Team,

Thank you for reviewing and accepting our manuscript, “MS: 1607386739368951 - Availability and Quality of Paraffin Blocks Identified by the Shared Pathology Informatics Network (SPIN): A Multi-institutional Study”.

We have made the final formatting changes requested. I thank you once again for your time and for publishing our work.

Sincerely,
-Ashok

Ashokkumar A. Patel
Formatting changes requested
-------------------------------
Please provide an affiliation for the author group or link it to an existing 
affiliation on the title page.

- Done. We have linked the SPIN group to the CDP/NCI affiliation already listed on the title page.

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.

- Done.

References - In the manuscript text, please place the reference before the full stop.

- Done.

Please title the figures section 'Figure legends'.

- Done.

Table 2 has been divided into a, b and c. It is important that tables are numbered in ascending numerical order - 2, 3 and 4 etc. Can we please ask that the tables are renumbered and any references to the tables in the text are updated to reflect these changes.

Please do the same for Table 3 and 5.

- Done. We have split all the tables that were previously grouped as individual tables and renumbered in ascending order. References to the tables in the text reflect the latest changes.